Author's response to reviews

Title: Accuracy of Visual Inspection with Acetic Acid as a Cervical Cancer Test Validated using Latent Class Analysis

Authors:

Lynne Gaffikin (earthlg@aol.com)
John MA McGrath (jmcmgrath@jhmi.edu)
Marc Arbyn (marbyn@iph.fgov.be)
Paul D Blumenthal (pdblumen@gmail.com)

Version: 2 Date: 13 January 2007

Author's response to reviews:

This version contains a list of abbreviations which was missing from the previous version plus some minor modifications to the acknowledgment section to indicate additional sources of funding for the time spent on writing the manuscript of one of the authors.

Additionally, I submitted an email shortly after uploading the manuscript the first time to indicate that I am willing to pay the processing charge however I found out from the website that Johns Hopkins University is a member. Three of the four authors at the time of the study were Johns Hopkins University staff/faculty and the organization supporting the research and manuscript development, JHPIEGO, is a Johns-Hopkins University affiliate. I do not know if this qualifies us for the member discount or not so please advise.